Author's response to reviews

Title: Influence of muscle mass and bone mass on the mobility of elderly women: an observational study

Authors:

Gláucia R Falsarella (grfalsarella@ig.com.br)
Ibsen B Coimbra (coimbra@fcm.unicamp.br)
Caroline C Barcelos (ccbarchel@gmail.com)
Isabele Iartelli (isabele.fisio@gmail.com)
Kedma T Montedori (kedmalopes@gmail.com)
Manuela NJ Santos (manuelanassin@gmail.com)
Anita L Neri (anitalbn@uol.com.br)
Arlete MV Coimbra (acoimbra@fcm.unicamp.br)

Version: 4  Date: 9 December 2013

Author's response to reviews: see over
Authors:
Gláucia R Falsarella, Ibsen B Coimbra, Caroline C Barcelos,
Isabele Iartelli, Kedma T Mantedori, Manuela NJ Santos,
Anita L Neri, Arlete M V Coimbra

Conflict of interest statement
The authors declare that they have no competing interests.

Corresponding Author: Gláucia Regina Falsarella
Address correspondence: Faculty of Medical Sciences, Family Health Program,
State University of Campinas (UNICAMP). Rua Vital Brasil, 50, Barão Geraldo,
Campinas-SP 13083-888, Brazil / telephone: 55 19 3521-8945
E-mail: grfalsarella@ig.com.br.

REVISED MANUSCRIPT
POINT-BY-POINT DESCRIPTION OF THE CHANGES MADE.

Reviewer: Hyehyung Shin
Revision 1.
Author: We have included in the Discussion (limitation of the study):
"Furthermore, were included elderly with chronic diseases, among them some
that can affect body composition, such as thyroid."
Revision 2.

Author: We have included in the Statistical Analysis: In this study, we used the multivariate logistic regression analysis to identify factors associated with mobility (gait speed and TUG). In this analysis, we considered the following variables: age, sarcopenia, bone mass, BMI, comorbidities and medication (current table 5 e 6).

We have included the analysis of covariance (ANCOVA) adjusted for age and comorbidities (current table 3 and 4).